Author's response to reviews

Title: Sugar-added beverages consumption among kindergarten children of Crete: effects on nutritional status and risk of obesity

Authors:

Manolis Linardakis (linman@med.uoc.gr)
Katerina Sarri (katsarri@med.uoc.gr)
Maria-Styliani Pateraki (melin_pat@yahoo.com)
Manolis Sbokos (manosmpok@yahoo.gr)
Anthony Kafatos (kafatos@med.uoc.gr)

Version: 6 Date: 4 August 2008

Author's response to reviews:

All formatting changes requested have been made (figure size corrected, line numbers deleted, keywords from abstract are removed, highlighting has been removed, addition to authors' contributions).